Author's response to reviews

Title: Impact of physical activity on energy balance, food intake and choice in normal weight and obese children in the setting of acute social stress: A randomized controlled trial

Authors:

Antje Horsch (antje.horsch@chuv.ch)
Marion Wobmann (marion.wobmann@unil.ch)
Susi Kriemler (susi.kriemler@ifspm.uzh.ch)
Simone Munsch (simone.munsch@unifr.ch)
Sylvie Borloz (sylvie.borloz-deladoey@chuv.ch)
Alexandra Balz (alexandra.balz@chuv.ch)
Pedro Marques-Vidal (pedro-manuel.marques-vidal@chuv.ch)
Ayala Borghini (ayala.borghini@chuv.ch)
Jardena J Puder (jardena.puder@chuv.ch)

Version: 2
Date: 8 April 2014

Author's response to reviews:

Dear Editor,

Many thanks for your letter dated 07.04.2014. We have added line and page numbers, as requested.

Sincerely,

Antje Horsch.